Author's response to reviews

Title: Role of Community Based Savings Groups (CBSGs) enhancing the utilization of Community Midwives in Chitral District of Pakistan.

Authors:

Qayyum Ali Noorani (qayyum.noorani@yahoo.com)
Iqbal Azam (iqbal.azam@aku.edu)
Babar Tasneem Shaikh (Shaikh.Babar@gmail.com)
Tharanga Ranasinghe (Tharanga.Ranasinghe@akdn.org)
Shazia Abbas (shazia.khoja@gmail.com)
Shakeela Wali (shakeela.raf@gmail.com)
Paul Rippey (paulrippeypdx@gmail.com)
Wajiha Javed (wajihajaved@hotmail.com)

Version: 2 Date: 27 September 2013

Author's response to reviews:

Dear Editor

According to the comments we received, we have revised the manuscript with the following specific changes.

1) Formatting

2) English language editing

3) Table and figure included in the manuscript after the reference section.

Thanking you in anticipation

Qayyum Ali Noorani